Author's response to reviews

Title: Measuring benefits and patients' satisfaction when glasses are not needed after cataract and presbyopia surgery: scoring and psychometric validation of the Freedom from Glasses Value Scale (FGVS(c))

Authors:

Gilles Berdeaux (Gilles.Berdeaux@AlconLabs.com)
Juliette Meunier (juliette.meunier@mapivalues.com)
Benoit Arnould (benoit.arnould@mapivalues.com)
Muriel Viala-Danten (muriel.viala@mapivalues.com)

Version: 5 Date: 19 May 2010

Author's response to reviews:

Dear Editor,

Please find enclosed our revised manuscript.

All required formatting changes (major and minor) have been made in this new version of the manuscript. A typo has been corrected in Table 5 and it induced a minor modification in the text, on page 10.

We hope that the revisions meet your expectations.

Yours sincerely,

Juliette Meunier, on behalf of the co-authors.